Author’s response to reviews

Title: Quick assessment with controlled attenuation parameter for hepatic steatosis in children based on MRI-PDFF as the gold standard

Authors:

Jaeseung Shin (drshinjs@yuhs.ac)
Myung-Joon Kim (mjkim@yuhs.ac)
Hyun Joo Shin (lamer-22@yuhs.ac)
Haesung Yoon (saydivine@yuhs.ac)
Seung Kim (pedks@yuhs.ac)
Hong Koh (khong@yuhs.ac)
Mi-Jung Lee (mjl1213@yuhs.ac)

Version: 3 Date: 28 Mar 2019

Author’s response to reviews:

Point-by-point response to the reviewer’s comments

Dear Editor,

We truly appreciate the Reviewers and the Editorial Board for their careful critique and helpful comments on our manuscript. We revised the manuscript in accordance with the suggestions and comments of the Reviewers. Detailed responses to each of the comments are as follows:

Technical Comments:

1. In the “Ethics approval and consent to participate” section of the Declarations, please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

   (On page 16, line 342-343) We modified the sentence as "This retrospective study was approved by the Institutional Review Board of Severance Hospital (reference number of 1-2016-0060)."
We also reviewed the Declarations section and confirmed once again that everything is included and described.

Again, we assure that this article has not been published elsewhere and is not being processed elsewhere.

Thank you for potential acceptance of our manuscript in BMC Pediatrics.

Best regards,

Mi-Jung Lee